Author's response to reviews

Title: Efficacy of methylsulfonylmethane supplementation on osteoarthritis of the knee: a randomized controlled study

Authors:

Eytan M Debbi (eytan.debbi@aya.yale.edu)
Gabriel Agar (gabyagar@gmail.com)
Gil Fichman (gil_fichman@hotmail.com)
Yaron Bar Ziv (ybarziv@013.net.il)
Rami Kardosh (kardosh27@hotmail.com)
Nahum Halperin (nhalpern@post.tau.ac.il)
Avi Elbaz (avie@apos.co.il)
Yiftah Beer (yiftah.beer@gmail.com)
Ronen Debi (debbi.ronen@gmail.com)

Version: 7 Date: 20 June 2011

Author's response to reviews: see over
Ronen Debi, MD
Department of Orthopedics
Assaf Harofeh Medical Center
Zerifin 70600, Israel

BMC Complementary and Alternative Medicine
Editorial Production Team
Dear Sir\Madam,

Thank you for reviewing the formatting of our paper, entitled, “Efficacy of methylsulfonylmethane supplementation on osteoarthritis of the knee: a randomized controlled study” [MS: 1312127821487565].

Upon reviewing the manuscript one final time we found one typographical error in the legend of Table 2 (page 28). We replaced “0 mm being the worst” with “100 mm being the worst” on the fifth to last line, and replaced “100 being the worst pain” with “0 being the worst pain” on the third to last line.

We thank you in advance for your patience and understanding.

Sincerely yours,

Ronen Debi, MD